Author’s response to reviews

Title: Optimal INR Level for Warfarin Therapy after Mechanical Mitral Valve Replacement

Authors:
Itthidet Kamthornthanakarn (dr.itthi@hotmail.com)
Rungroj Krittayaphong (rungroj.kri@mahidol.ac.th)

Version: 3 Date: 01 Apr 2019

Author’s response to reviews:

Response to comments from reviewers and editors
BCAR-D-18-00341R2
Optimal INR Level for Warfarin Therapy after Prosthetic Mitral Valve Replacement
Itthidet Kamthornthanakarn, MD; Rungroj Krittayaphong, M.D.
BMC Cardiovascular Disorders

Editor Comments:

1. If the need for consent was waived by an ethics committee or is deemed unnecessary according to national regulations, please clearly state this, including the name of the committee or a reference to the relevant legislation. Further, please clarify whether any administrative permissions were required to access the raw data. If so please state who granted permission in the Ethics approval and consent to participate section.

Response:
We replaced the statement under Ethic approval in the methods section and in the Ethics approval and consent to participate section with the following statement ‘The protocol for this study was approved by the Siriraj Institutional Review Board (SIRB), Faculty of Medicine Siriraj Hospital, Mahidol University, Bangkok, Thailand. The Institutional Review Board of Faculty of Medicine Siriraj Hospital approved the study after determining that the study met the definition of research not requiring informed consent.’ (page 4 line 22-25 and page 12 line 12-15)

2. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it
in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

Response:
We replaced the statement under the Availability of Data Materials with the following statement ‘The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.’(page 12 line 21-22)